Author’s response to reviews

Title: Factors associated with the impact of quality improvement collaboratives in mental healthcare: an exploratory study

Authors:

Marleen H. Versteeg (marleen.versteeg@hetnet.nl)
Miranda G.H. Laurant (M.Laurant@iq.umcn.nl)
Gerdien C. Franx (gfranx@trimbos.nl)
Annelies J. Jacobs (J.Jacobs@iq.umcn.nl)
Michel J.P. Wensing (M.Wensing@iq.umcn.nl)

Version: 9 Date: 27 December 2011

Author’s response to reviews: see over
Dear Prof Foy,

We are pleased to submit the final version of the manuscript entitled ‘Factors associated with the impact of quality improvement collaboratives in mental healthcare: an exploratory study’. We submitted our version by Internet.

We processed your final remarks:

- Author list: we wrote the list of authors in the manuscript exactly as they are in the submission system, both in style and order.
- Tables: we changed the order of the Tables in the manuscript and provided them in the same order at the end of the manuscript.
- Highlighting/tracking: we removed all highlighting and tracking from the manuscript.
- References: we replaced the term 'et al.' in the reference list by the names of the other authors.
- Tables: we removed the visible vertical lines from our tables.

Every care has been taken to improve the manuscript with your advice. We look forward to your response.

Yours sincerely, on behalf of the other authors,

Marleen Versteeg
(M.H.Versteeg, MSc, Junior researcher)

Correspondence details:
M.G.H. Laurant, PhD, Senior research fellow
Scientific Institute for Quality of Healthcare (IQ healthcare)
Radboud University Nijmegen Medical Centre
P.O. Box 9101, 114 IQ healthcare
6500 HB Nijmegen
The Netherlands
T 0031243615305
F 0031243540166

Email address:
ML M.Laurant@iq.umcn.nl

Authors:
Marleen H. Versteeg¹, Miranda G.H. Laurant², Gerdien C. Franx³, Annelies J. Jacobs⁴, Michel J. P. Wensing⁵

¹⁴⁻⁵ Scientific Institute for Quality of Healthcare (IQ healthcare), Radboud University Nijmegen Medical Centre, P.O. Box 9101, 114 IQ healthcare, 6500 HB Nijmegen
³ G.C. Franx, MSc, Program coordinator, Trimbos-instituut Da Costakade 45, 3521 VS Utrecht